Author's response to reviews

Title: Protocol for Northern Ireland Caries Prevention in Practice Trial (NIC-PIP) trial: a randomised controlled trial to measure the effects and costs of a dental caries prevention regime for young children attending primary care dental services

Authors:

Martin Tickle (martin.tickle@manchester.ac.uk)
Helen V Worthington (Helen.Worthington@manchester.ac.uk)
Michael Donaldson (Michael.Donaldson@hscni.net)
Seamus Killough (sbkilldent@hotmail.com)
Ciaran O'Neill (ciaran.oneill@nuigalway.ie)
Grainne Crealey (Grainne.crealey@crsc.n-i.nhs.uk)
Matt Sutton (matt.sutton@manchester.ac.uk)
Solveig Noble (Solveig.Noble@northerntrust.hscni.net)
Margaret Greer (MargaretGreer@crsc.n-i.nhs.uk)
Milsom Keith (keith.milsom@hsthpct.nhs.uk)

Version: 3 Date: 23 September 2011

Author's response to reviews:

Requested revisions have been made.